Author's response to reviews

Title: Atypical atrial myxomas in two asymptomatic patients. A case report.

Authors:

Nicholas A Charokopos (charokoposnick@hotmail.com)
Efthymia Rouska (charokoposnick@hotmail.com)
Christodoulos Pliakos (pliakos@med.auth.gr)
Efstathios D Pagourelias (statpag@yahoo.gr)
Christoforos Foroulis (charokoposnick@hotmail.com)
Nikolaos Papadopoulos (charokoposnick@hotmail.com)

Version: 2 Date: 1 September 2009

Author's response to reviews: see over
Dear Editor,
We would like to thank you and your reviewer for your useful advice. We tried hard to embody all your reviewer’s suggestions in our revised manuscript. Please, find below a point by point response to reviewer’s comments.
Thank you very much.
Best Regards

Nicholas Charokopos
Corresponding author

Reviewer’s report:
This is the description of two cases unusual location for atrial myxoma. The cases are well described and interesting for the readership of the journal.
Thank you for your kind remarks

The discussion is a bit too long.
We have thoroughly revised and shortened our discussion section.

Please upload also clips (there is no space limit).
Unfortunately due to the very bad quality of stored echo-clips is not possible to edit and upload them. Additionally, we can not convert them in a format compatible to the journal's style.